Author's response to reviews

Title: Socio-Epidemiological Features of 2002 Plague Outbreak in Himachal Pradesh, India: A Qualitative Study

Authors:

Sonu Goel (sonugoel007@gmail.com)
Harvinder Kaur (harvinderkaur315@gmail.com)
A K Gupta (anil1031@hotmail.com)
Umesh S Chauhan (uschauhan78@gmail.com)
Amarjeet Singh (amarminhas56@rediffmail.com)

Version: 7 Date: 27 March 2014

Author's response to reviews: see over
Dear Editor,

We thank the reviewers for useful comments on the article. We think that the article has been improved after incorporation of their suggestions. We have carefully looked at the article for any grammatical and technical errors and also sought assistance from expert in English Language. We hope that the article shall now be published and the important findings are disseminated to wider global audience. The article provides in-depth socio-epidemiological details of plague outbreak in India, which has been untouched so far by other researchers.

The manuscript has been read and approved by all the authors, and the requirements for authorship have been met, and that each author believes that the manuscript represents honest work. The paper follows the guidelines of the journal. It has not been published elsewhere and is not under consideration anywhere else. There is no financial support for the said article. The authors declare no conflict of interest.

Thanking you.

Dr. Sonu Goel, (Corresponding Author), Assistant Professor, Department of Community Medicine, School Of Public Health, Post Graduate Institute of Medical Education and Research, Sector: 12, Chandigarh – 160012, India. Ph:+91-9914208027, email: sonugoel007@yahoo.co.in